

Limnology and Oceanography Letters 5, 2020, 84–91 © 2020 The Authors. Limnology and Oceanography published by Wiley Periodicals, Inc. on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/lol2.10143

SPECIAL ISSUE-LETTER

Feeding behavior is the main driver for microparticle intake in mangrove crabs

Christelle Not^{1,2}* Cheuk Yan Iris Lui,^{1,3} Stefano Cannicci^{2,3,4}

¹Department of Earth Sciences, University of Hong Kong, Pokfulam, Hong Kong; ²Swire Institute of Marine Science, University of Hong Kong, Cape d'Aguilar, Shek O, Hong Kong; ³School of Biological Sciences, University of Hong Kong, Pokfulam, Hong Kong; ⁴Department of Biology, University of Florence, Sesto Fiorentino, Italy

Scientific Significance Statement

Microplastics, which are plastic pieces smaller than 5 mm, are commonly found in the marine environment, where they are available for ingestion by marine fauna. Because of both their root structure and their geographic distribution, mangrove forests are known as sinks of microplastics, but very little is known about the factors controlling the microplastics ingestion by mangrove associated crabs, which are a dominant and ecologically relevant ecosystem component. We observed that both the level of plastic contamination in the mangrove and the feeding habit of the crab's species play critical roles in the abundance and type of microplastic ingested by mangrove crabs.

Abstract

As marine plastic debris is primarily sourced from terrestrial input, coastal environments are particularly affected by deposition. Because of their pneumatophores, mangroves have been recognized for their importance in confining plastic waste. Crabs are a dominant component of the mangrove ecosystem and play a critical role in maintaining healthy and resilient mangrove forests. Therefore, the presence of debris fragmented from waste, in their habitat is a potential threat. However, the potential ingestion of microplastic pieces by mangrove crabs has not yet been investigated. Here, we quantified microparticles found in the cardiac stomachs and gill chambers of four species of crabs. All specimens collected had anthropogenic microparticles present either via their digestive or respiratory systems. We observed significant variability in the abundance and types of anthropogenic microparticles across sites and species. Interspecific differences appear to be explained by their particular feeding habits, with less selective species ingesting more particles.

Data Availability Statement: Data will be made available in the dryad data repository at the final submission of the manuscript.

Associate editor: Elise Granek

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

This article is an invited paper to the Special Issue: Microplastics in marine and freshwater organisms: Presence and potential effects Edited by: Dr Elise Granek, Portland State University, Dr Susanne Brander, Oregon State University, and Dr Erika Holland, California State University, Long Beach

^{*}Correspondence: cnot@hku.hk

Author Contribution Statement: C.N., C.Y.I.L., and S.C. designed the study. C.Y.I.L. carried out the experiment and conducted a preliminary analysis of the data with the help of C.N. S.C. performed statistical analyses, and C.N. and S.C. wrote the manuscript.

Marine debris, and in particular plastic debris, has been observed throughout large ranges of marine environments (Moore et al. 2001; Widmer and Hennemann 2010; Galgani et al. 2015). Rivers have been identified as the main sources of plastic litter, which results in considerable pollution in coastal regions. Hong Kong coastlines are no exception, as Hong Kong is located at the mouth of the Pearl River, the third largest riverine source of plastic into the ocean (Lebreton et al. 2017). Large plastic debris can be easily observed stranded on many coastal habitats of Hong Kong, but investigation of their abundance and composition is mostly limited to sandy beaches, with other coastal habitats remaining understudied (Cheung et al. 2018b; Ho and Not 2019). Macroplastics are fragilized by several environmental variables, including UV radiation, humidity, and wave action. As such, this debris is susceptible to fragmentation, eventually leading to the formation of microplastics (i.e., plastic pieces smaller than 5 mm; Cole et al. 2011). While microplastics have been found in all marine environments, coastal environments are particularly affected due to their typical proximities to river deltas and urbanized regions (Peters and Bratton 2016; Lebreton et al. 2017; Chan et al. 2019). In Hong Kong, microplastics have been identified in coastal surface waters (Cheung et al. 2018a; So et al. 2018), sandy beaches (Fok and Cheung 2015; Fok et al. 2017), marine sediments (Tsang et al. 2017), and also within marine organisms (Chan et al. 2019).

Because of their coastal habitats, mangroves are highly susceptible to debris exposure. Furthermore, the spatial complexity characteristic of mangrove ecosystems provides many opportunities for debris to become entangled, which enhances their role as sinks for plastic pollution (Garcés-Ordóñez et al. 2019; Martin et al. 2019). Macroplastics trapped within coastal forests are easily fragmented into microplastics, and recent studies have reported a high abundance of microplastic accumulation within mangrove roots systems (Garcés-Ordóñez et al. 2019; Li et al. 2019; Martin et al. 2019). With the exception of the abovementioned studies, there are still very few estimates of macro- and microplastic abundances within mangroves. These forests host diverse and highly specialized invertebrate fauna (Duke et al. 2007; Cannicci et al. 2008), which play critical roles in maintaining ecosystem functions and could be strongly impacted by plastic pollution.

Hong Kong mangroves are characterized by a high density and diversity of crabs and molluscs (Lee 2000). These two taxa dominate the mangrove food web (Kristensen et al. 2017), and their activity has a significant engineering effect on the sediment and whole-forest functionality (Kristensen 2008). Although recent experimental evidence has emphasized the important role played by these bioengineers in maintaining healthy and functional mangrove forests, very few studies have investigated the impact of anthropogenic stressors (Cannicci et al. 2009), such as plastic pollution, on their physiology. However, based on experimental tests, crabs have been shown to uptake microplastic via dietary input and ventilation, leading to a reduction in their food consumption and, consequently, of their energy budget (Watts et al. 2015, 2016). In addition, transfer of microplastics from stomachs to hepatopancreas has been identified in the tropical fiddler crab, *Uca rapax* (Brennecke et al. 2015). Observations of microplastic ingestion by crabs in natural environments are, however, still limited (Wójcik-Fudalewska et al. 2016; Welden et al. 2018).

Over the last decades, mangroves have been under multiple threats and are disappearing worldwide at the alarming rate of 1-2% per year (Duke et al. 2007; Richards and Friess 2015; Friess et al. 2019), especially along the coast of the South China Sea. Hong Kong mangroves, however, have proved to be truly resilient, since about 60 mangrove stands, covering an area of about 500 ha, are still thriving along the coast (Tam et al. 1997). These remaining pockets of forest still provide paramount ecosystem services and functions (Duke et al. 2007). They support pelagic and benthic marine food webs, evidenced through their significant contributions of organic carbon (Meynecke et al. 2007), but also act as carbon sinks and thus mitigate climate change (Donato et al. 2011). Mangrove forests further reduce coastal erosion and act as filters for pollutants, which are trapped within their fine sediment and are not dispersed into open waters (MacFarlane et al. 2007). Due to these important ecological roles and the increasing threat of plastic litter, it is crucial to characterize the types and ecological implications of plastic litter within mangrove habitats. Here, we use a field approach to look at the ingestion of anthropogenic particles by four mangrove crab species occupying different levels of the food web. The transfer of anthropogenic debris as microplastic, from the environment into crab organs, has been primarily observed via feeding and gill ventilation. Therefore, we investigated the presence of anthropogenic debris in the cardiac stomach and gills of mangrove crabs, to better understand and characterize the main intake processes. Potential transfer from stomach to hepatopancreas has been suggested in experimental studies (Brennecke et al. 2015) but it is beyond the scope of this study, which focuses on intake processes rather than transfer among internal organs. Here, we concentrate on possible differences in intake among crab populations subject to different levels of plastic pollution and among crab species characterized by different feeding habits to test whether microdebris availability and feeding behavior affect the microplastic intake in these keystone species in mangrove habitats.

Materials and methods

Crabs were collected between October 21st, 2017 and November 3rd, 2017 from three mangrove forests in Hong Kong: (1) Ha Pak Nai (HPN, 14 individuals sampled), located in the north west of Hong Kong, (2) Pak Tam Chung (PTC, 19 individuals sampled), and (3) Yung Shue O (YSO, 16 individuals sampled), both located on the east coast (Fig. 1; Table 1).

The three sites are small patches (< 1 ha) of mangrove forests dominated by Kandelia obovata trees and located at the mouth of small rivers (Tam et al. 1997). However, HPN, situated within the estuary of the Pearl River, is characterized by higher salinity and nutrients compared to PTC and YSO (Duprey et al. 2016). Within each mangrove site, adult crabs, that is, morphologically and sexually mature individuals, belonging to four crab species were collected by hand during their low tide activity period by two expert researchers. The focal species were: a detritivorous sesarmid, Parasesarma bidens (n = 15, size range 13 mm < carapace width [CW], > 20 mm), known to ingest mainly plant material and organic sediment on the mangrove floor (Poon et al. 2010); the grapsid *Metopograpsus frontalis* $(n = 14, \dots, n)$ size range 15 mm < CW > 20 mm), which is an opportunistic feeder able both to predate on small invertebrates and to feed on plant material (Fratini et al. 2000; Poon et al. 2010); the ocypodid Paraleptuca splendida (n = 13, size range 13 mm < CW > 18 mm), which is known as a floating feeder,



Fig. 1. Sampled mangrove sites within Hong Kong. *n* represents the number of crabs sampled.

since it places a small amount of substratum in its buccal cavity and sorts food particles by passing water through the buccal cavity (Crane 1975), and the predator portunid Thalamita crenata (n = 7, size range 25 mm < CW > 35 mm), which ambushes crabs and searches for bivalves on the mud flat (Cannicci et al. 1996). Although all of the above four species show some degree of terrestrialization and are active at low tide, physiological and morphological differences are evident. Parasesarma bidens, M. frontalis, and T. crenata strongly rely on gills to breath and have evolved various mechanisms to retain water into their gill chambers (Little 2009). On the other hand, Paraleptuca splendida, like most fiddler crabs, evolved a true lung within the gill chamber and mostly uses its gills for osmoregulation and excretion (Paoli et al. 2015). Due to the strong differences in abundances shown by the above species at the study sites, we could not obtain a balanced design, although we could standardize for sizes and sex. To minimize stress, the collected crabs were transported in buckets with leaves and sediments, rinsed with ultra-pure water (Milli-Q) to remove sediment from the surface of their carapaces. Finally, crabs were stored at -18°C until further processing.

Stomachs and gills were dissected and digested by a 10-min sonication treatment followed by a hydrogen peroxide (35%) bath heated at 80°C for 2 h. The digested solution was filtered through a 0.3 μ m pore size glass filter and dried. Finally, 0.5 mL of 1% Rose Bengal was dropped on the filter to facilitate the identification of remaining organic material. Each filter was examined under a stereomicroscope (Carl Zeiss Stemi 305) and particles $\geq 10 \ \mu m$ were identified, categorized, and counted as fragment, fiber, pellet, and bead. Fragments were categorized as angular-shaped pieces, fibers as elongate cylindric pieces, pellets as oval-shaped pieces, and beads as spherical particles (Supporting Information Fig. S1). Because categorical distinctions of particles smaller than 10 μ m were difficult, especially between bead and pellet, and because such particles are unidentifiable via fourier transform infrared spectroscopy (FTIR) analysis, we limited our investigation to particles bigger than 10 μ m. Finally, selected particles, based on their sizes and shapes, were analyzed with a Bruker Lumos

Table 1. Number of microparticles found in gills and stomachs by crab species and by location and number of crab individuals analyzed for each species and location. Notes: Nb microparticles refers to the total number of particles found across all crabs investigated for a particular species and location. Nb ind. refers the number of individual crabs analyzed.

Species/sites	YSO		РТС		HPN		Total	
	Nb microparticles	Nb ind.	Nb microparticles	Nb ind.	Nb microparticles	Nb ind.	Nb microparticles	Nb ind.
Parasesarma bidens	365	3	393	6	615	6	1373	15
Paraleptuca splendida	86	6	53	4	194	3	333	13
M. frontalis	534	5	197	6	238	3	969	14
T. crenata	49	2	99	3	143	2	291	7
Total	1034	16	742	19	1190	14	2966	49

FTIR microscope using the attenuated reflectance transmission attenuated total reflectance fourier transform infrared spectroscopy (ATR-FTIR) method to identify their compositions. Spectra were identified using the Bruker spectrum library. To reduce potential contamination, all equipment were rinsed with MilliQ and methanol, all experiments were run under a fume hood, unnecessary opening of the filtration setup and petri dish were avoided, and cotton laboratory coats were used. Additionally, a procedural blank was used during each set of samples, with a total of nine replicates.

The total amount of recorded particles and the different amount of particle types found were analyzed with univariate and multivariate permutational analysis of variance (PERMANOVA, Anderson et al. 2008) designs, respectively. Both analyses were applied to test for differences in microplastic presence across locations (three levels, fixed and orthogonal), species (four levels, fixed and orthogonal), and between gills and stomach (fixed and orthogonal). In particular, the total amount of particles found, and the overall composition of particle types present were analyzed with univariate and multivariate PERMANOVA designs, respectively. To standardize for possible relationships between the specimen size and the amount of bodily microdebris found, the number of particles belonging to each of the selected categories found in each specimen was divided by their CW, a standard measure of size for crabs. Due to their heteroscedasticity assessed via PERMDISP, a multivariate version of the Leven test (Anderson et al. 2008), the data were logtransformed and the Euclidean distance was used to calculate the dissimilarity matrix. When appropriate, post hoc pairwise tests were performed to examine significant differences among levels of factors. A principal component analysis (PCA) on previously normalized data was performed as an unconstrained ordination to visualize patterns of particle composition in stomach and gills. To further analyze the differences in the number of particles of the various categories found in the stomach and gills of the different species analyzed, we performed a canonical analysis of principal coordinates (CAP). All analyses were based on 9999 permutations and were carried out using the software PRIMER 7 and its add-on package PERMANOVA+ (Anderson et al. 2008).

Results and discussion

All 49 crab specimens had particles present either in their stomach or in/on their gills, with only one individual containing particles exclusively in its stomach, reflecting the high contamination of mangrove crabs by particles. From the sampled specimens, we retrieved a total 2966 microparticle pieces, with almost 90% of the pieces present in the stomach and 10% in the gills, indicating that ingestion is the main driver of microparticle intake (df = 1; F = 75.1; p < 0.0001, three-way PERMANOVA, Table 1, Supporting Information Table S1). Based on the method used for microparticles recovery from gills (dissection and then digestion), it is not possible to

distinguish particles that were trapped at the surface vs. the ones inside the gills; however, only 10% of the microparticles retrieved where found in the gills, suggesting that the ventilation system is not a main path of microparticles intake, at least for microparticles bigger than 10 μ m. No significant correlation was observed between the content of microparticles recorded in stomach, gills, or both and the size of the individual crab, both measured as carapace width and length. An average of two microparticles was observed in procedural blanks and was thus used to correct the number of microparticles in crabs. We tested 46 pieces with ATR-FTIR but could only confirm the composition of nine of them. All of the other pieces were either too small for signal-to-background differentiation, or the presence of rose Bengal or remaining organic matter altered the readability of the spectrum. Of these nine particles, one fiber was identified as polyethylene (PE), one fragment as polyethylene terephthalate (PET), four particles as rayon, two particles as paint, and one as cobalt. Based on these results, we ascertained that most of the microparticles identified were derived from human-made material (89%, considering that cobalt may come from a natural source). If a majority of these microparticles can be considered as originating from human material, only 22% are identified as plastic polymer (PE and PET). However, this is based on a limited number of analyses and we recognize the need for additional investigations to further verify the abundance of microplastics in mangrove crab organs. Therefore, we use the term "microparticles" here to describe the combination of plastic and nonplastic micropieces from anthropogenic sources.

Both the abundance and composition of microparticle types found in crabs significantly vary between species (df = 3; *F* = 5.4; *p* < 0.01 and df = 3; *F* = 9.0; *p* < 0.0001 for abundance and composition, respectively, three-way PERMANOVA) and mangrove sites from which they were collected (df = 2; F = 3.4; p < 0.05 and df = 2; F = 4.6; p < 0.01 for abundance and composition, respectively, three-way PERMANOVA; Fig. 2). Crabs from the HPN mangrove located on the west coast of Hong Kong were shown to have significantly higher abundances of microparticles compared to crabs from the two other mangroves (HPN > PTC = YSO, PERMANOVA post hoc tests). A distinction in microplastic content in Hong Kong waters between the east and west side of the region has been previously shown to be linked to the influence of the Pearl River (Fok and Cheung 2015). Since our sampling sites do not differ in terms of structural characteristics, tree dominance, or area, the east-west difference observed in the abundance of microparticles in crab organs is likely linked to the influence of the Pearl River.

We also observed a significant difference in the abundance of microparticles ingested by different species, with *Parasesarma bidens* and *M. frontalis* characterized by similar amounts of microdebris (t = 1.16, p = 0.25, PERMANOVA post hoc test), and with both species higher than the ocypodid



Fig. 2. Box plot showing the abundance microparticles found in gills and stomach of individual mangrove crabs at different locations (**A**) and in different species across locations (**B**). Sites are YSO, PTC, both located in the east side of Hong Kong, and HPN located in the north west of Hong Kong. The number of collected specimens per location and per species is shown below the bar. The box plots represent the nonparametric parameters, such as medians, quartiles, and 95% ranges.

Paraleptuca splendida (t = 5.43, p < 0.001 and t = 3.92, p < 0.01, for Parasesarma bidens and M. frontalis, respectively, PER-MANOVA post hoc tests). The predator T. crenata had a lower amount of ingested microparticles compared to the omnivorous Parasesarma bidens (t = 2.50, p < 0.03, PER-MANOVA post hoc test), but a higher amount than the microphytobenthos feeder Paraleptuca splendida (t = 2.26, p < 0.05,PERMANOVA post hoc test). These results can be easily explained by the food preferences and the feeding habits of these species. The highest rates of microparticles were recorded in the stomachs of the detritivore Parasesarma bidens, known to actively pick up plant materials and particulate organic matter (POM) from the sediment (Lee 2000; Poon et al. 2010), and of the omnivorous M. frontalis, known to ingest similar rates of animal, plant, and POM from the sediment (Poon et al. 2010). T. crenata is also a generalist feeder, but this crab avoids feeding on sediment and exerts a strong preference for predation on crabs and molluscs (Cannicci et al. 1996). Finally, out of the four species, the lowest content of microparticles was found in the stomachs of Paraleptuca splendida, which, like other fiddler

crabs, filters pellets of mud to feed on microphytobenthos, using water stored in its gill chambers (Icely and Jones 1978; Dye and Lasiak 1987; Kawaida et al. 2019). Such active filtering could prevent this species from ingesting the heaviest microparticles, which are instead discharged into the pseudo-pellets contained within inorganic components of the mud.

PCA ordination results show clear patterns regarding the types of microparticles found in stomachs and gills across the different species (Fig. 3A). CAP showed that stomachs contained more fragments and beads, whereas gills contained mainly fibers (Fig. 3A,B). These results confirm that both dietary and ventilation processes are responsible for microparticle intake by crabs (Watts et al. 2015, 2016). The preferential presence of fibers during the ventilation process is consistent with microparticles found in filtering organisms such as oysters and mussels (Li et al. 2016, 2018). CAP also confirmed that *Parasesarma bidens* and *M. frontalis* preferentially ingested fragments (~ 80%) and beads (~ 13%), whereas *T. crenata* ingested a mixture of fragments, beads, and fibers and *Paraleptuca splendida* ingested mostly fibers (Fig. 3B). Based



Fig. 3. Two-dimensional scatter plots of the first and second principal components (**A**) and of the first and second canonical axes (**B**) of microparticle type compositions found in the stomach and gill chambers of four species of mangrove crab from Hong Kong. Open symbols represent the data collected from the gills, closed ones the data from the gut. Triangles represent data from YSO, dots represent PTC, and squares represent HPN. Vectors of the linear correlations between individual variables are superimposed on the graph.

on these observations, we suggest that both the abundance and type of microparticles ingested in the cardiac stomach depend on the feeding habits of the crabs. For example, *Parasesarma bidens* as a detritivore ingested predominantly fragments whereas *Paraleptuca splendida* ingested less fragments but more fibers, which have lower sinking velocity (Khatmullina and Isachenko 2017; Hoellein et al. 2019) and could be ingested together with microphytoplankton during the filtering process carried out by this species.

Conclusion

Hong Kong mangroves are highly polluted by human activities, and mangrove crabs, one of the keystone taxa in the mangrove environment, appear to be heavily impacted by such anthropogenic debris. All of the sampled specimens in this study, belonging to four dominant species and located in three different mangroves throughout Hong Kong, present intake of microparticles. We have shown here that the abundances and types of microparticles found in crab stomachs are related to the crab's role in the food web and their feeding habits, with crabs that employ generalist feeding strategies ingesting a higher abundance and more diverse range of types of microparticles. Although we could confirm that crabs absorbed microplastics via both their ingestion and ventilation efforts, we clearly showed that most of the intake happens via feeding.

References

- Anderson, M. J., R. N. Godley, and K. R. Clarke. 2008, *PER-MANOVA+ for PRIMER: Guide to software and statistical methods*. PRIMER-E Ltd.
- Brennecke, D., E. C. Ferreira, T. M. M. Costa, D. Appel, B. A. P. da Gama, and M. Lenz. 2015. Ingested microplastics

(>100μm) are translocated to organs of the tropical fiddler crab *Uca rapax*. Mar. Pollut. Bull. **96**: 491–495. doi:10.1016/j.marpolbul.2015.05.001

- Cannicci, S., F. Dahdouh-Guebas, D. Anyona, and M. Vannini. 1996. Natural diet and feeding habits of *Thalamita crenata* (Decapoda: Portunidae). J. Crustacean Biol. **16**: 678. doi:10.2307/1549188
- Cannicci, S., D. Burrows, S. Fratini, T. J. Smith, J. Offenberg, and F. Dahdouh-Guebas. 2008. Faunal impact on vegetation structure and ecosystem function in mangrove forests: A review. Aquat. Bot. 89: 186–200. doi:10.1016/j.aquabot. 2008.01.009
- Cannicci, S., and others. 2009. Effects of urban wastewater on crab and mollusc assemblages in equatorial and subtropical mangroves of East Africa. Estuar. Coast. Shelf Sci. **84**: 305–317. doi:10.1016/j.ecss.2009.04.021
- Chan, H. S. H., C. Dingle, and C. Not. 2019. Evidence for non-selective ingestion of microplastic in demersal fish. Mar. Pollut. Bull. **149**: 110523. doi:10.1016/j.marpolbul. 2019.110523
- Cheung, P. K., L. Fok, P. L. Hung, and L. T. O. Cheung. 2018*a*. Spatio-temporal comparison of neustonic microplastic density in Hong Kong waters under the influence of the Pearl River Estuary. Sci. Total Environ. **628–629**: 731–739. doi: 10.1016/j.scitotenv.2018.01.338
- Cheung, P. K., P. L. Hung, and L. Fok. 2018b. River microplastic contamination and dynamics upon a rainfall event in Hong Kong, China. Environ. Process. 6: 253–264. doi: 10.1007/s40710-018-0345-0
- Cole, M., P. Lindeque, C. Halsband, and T. S. Galloway. 2011. Microplastics as contaminants in the marine environment: A review. Mar. Pollut. Bull. 62: 2588–2597. doi:10.1016/j. marpolbul.2011.09.025
- Crane, J. 1975, *Fiddler crabs of the world. Ocypodidae: Genus* Uca. Princeton Univ. Press. doi:10.1002/iroh.19780630113
- Donato, D. C., J. B. Kauffman, D. Murdiyarso, S. Kurnianto, M. Stidham, and M. Kanninen. 2011. Mangroves among the most carbon-rich forests in the tropics. Nat. Geosci. 4: 293–297. doi:10.1038/ngeo1123
- Duke, N. C., and others. 2007. A world without mangroves? Science **317**: 41–42. doi:10.1126/science.317.5834.41b
- Duprey, N. N., M. Yasuhara, and D. M. Baker. 2016. Reefs of tomorrow: Eutrophication reduces coral biodiversity in an urbanized seascape. Glob. Chang. Biol. 22: 3550–3565. doi: 10.1111/gcb.13432
- Dye, A. H., and T. A. Lasiak. 1987. Assimilation efficiencies of fiddler crabs and deposit-feeding gastropods from tropical mangrove sediments. Comp. Biochem. Physiol. A Physiol. 87: 341–344. doi:10.1016/0300-9629(87)90132-0
- Fok, L., and P. K. Cheung. 2015. Hong Kong at the Pearl River Estuary: A hotspot of microplastic pollution. Mar. Pollut. Bull. 99: 112–118. doi:10.1016/j.marpolbul.2015.07.050
- Fok, L., P. K. Cheung, G. Tang, and W. C. Li. 2017. Size distribution of stranded small plastic debris on the coast of

Guangdong, South China. Environ. Pollut. **220**: 407–412. doi:10.1016/j.envpol.2016.09.079

- Fratini, S., S. Cannicci, L. M. Abincha, and M. Vannini. 2000. Feeding, temporal, and spatial preferences of *Metopograpsus thukuhar* (Decapoda; Grapsidae): An opportunistic mangrove dweller. J. Crustacean Biol. **20**: 326–333. doi:10. 1651/0278-0372(2000)020[0326:FTASPO]2.0.CO;2
- Friess, D. A., and others. 2019. The state of the world's mangrove forests: Past, present, and future. Annu. Rev. Environ. Resour.
 44: 1–27. doi:10.1146/annurev-environ-101718-033302
- Galgani, F., G. Hanke, and T. Maes. 2015. Global distribution, composition and abundance of marine litter, p. 29–56. *In Marine anthropogenic litter*. Springer, Cham.
- Garcés-Ordóñez, O., V. A. Castillo-Olaya, A. F. Granados-Briceño, L. M. Blandón García, and L. F. Espinosa Díaz. 2019. Marine litter and microplastic pollution on mangrove soils of the Ciénaga Grande de Santa Marta, Colombian Caribbean. Mar. Pollut. Bull. **145**: 455–462. doi:10. 1016/j.marpolbul.2019.06.058
- Ho, N. H. E., and C. Not. 2019. Selective accumulation of plastic debris at the breaking wave area of coastal waters. Environ. Pollut. 245: 702–710. doi:10.1016/j.envpol.2018.11.041
- Hoellein, T. J., A. J. Shogren, J. L. Tank, P. Risteca, and J. J. Kelly. 2019. Microplastic deposition velocity in streams follows patterns for naturally occurring allochthonous particles. Sci. Rep. 9: 1–11. doi:10.1038/ s41598-019-40126-3
- Icely, J. D., and D. Jones. 1978. Factors affecting the distribution of the genus *Uca* (Crustacea: Ocypodidae) on an East African shore. Estuar. Coast. Mar. Sci. 6: 315–325. doi:10. 1016/0302-3524(78)90019-1
- Kawaida, S., K. Nanjo, N. Ohtsuchi, H. Kohno, and M. Sano.
 2019. Cellulose digestion abilities determine the food utilization of mangrove estuarine crabs. Estuar. Coast. Shelf Sci.
 222: 43–52. doi:10.1016/j.ecss.2019.04.004
- Khatmullina, L., and I. Isachenko. 2017. Settling velocity of microplastic particles of regular shapes. Mar. Pollut. Bull. 114: 871–880. doi:10.1016/j.marpolbul.2016.11.024
- Kristensen, E. 2008. Mangrove crabs as ecosystem engineers; with emphasis on sediment processes. J. Sea Res. **59**: 30–43. doi:10.1016/j.seares.2007.05.004
- Kristensen, E., S. Y. Lee, P. Mangion, C. O. Quintana, and T. Valdemarsen. 2017. Trophic discrimination of stable isotopes and potential food source partitioning by leaf-eating crabs in mangrove environments. Limnol. Oceanogr. 62: 2097–2112. doi:10.1002/lno.10553
- Lebreton, L. C. M., J. Van Der Zwet, J. W. Damsteeg, B. Slat, A. Andrady, and J. Reisser. 2017. River plastic emissions to the world's oceans. Nat. Commun. 8: 1–10. doi:10.1038/ ncomms15611
- Lee, S. Y. 2000. Carbon dynamics of Deep Bay, eastern Pearl River Estuary, China. II: Trophic relationship based on carbon- and nitrogen-stable isotopes. Mar. Ecol. Prog. Ser. **205**: 1–10. doi:10.3354/meps205001

- Li, B., Z. Zang, Y. Yao, X. Zou, W. Wang, Y. Li, W. Yu, and T. Wang. 2018. Preliminary study of the source apportionment and diversity of microplastics: Taking floating microplastics in the South China Sea as an example. Environ. Pollut. 245: 965–974. doi:10.1016/j.envpol.2018.10.110
- Li, J., X. Qu, L. Su, W. Zhang, D. Yang, P. Kolandhasamy, D. Li, and H. Shi. 2016. Microplastics in mussels along the coastal waters of China. Environ. Pollut. **214**: 177–184. doi:10.1016/j.envpol.2016.04.012
- Li, R., L. Zhang, B. Xue, and Y. Wang. 2019. Abundance and characteristics of microplastics in the mangrove sediment of the semi-enclosed Maowei Sea of the South China Sea: New implications for location, rhizosphere, and sediment compositions. Environ. Pollut. **244**: 685–692. doi:10.1016/ j.envpol.2018.10.089

Little, C. 2009, The colonisation of land. Cambridge Univ. Press.

- MacFarlane, G. R., C. E. Koller, and S. P. Blomberg. 2007. Accumulation and partitioning of heavy metals in mangroves: A synthesis of field-based studies. Chemosphere **69**: 1454–1464. doi:10.1016/j.chemosphere.2007.04.059
- Martin, C., H. Almahasheer, and C. M. Duarte. 2019. Mangrove forests as traps for marine litter. Environ. Pollut. **247**: 499–508. doi:10.1016/j.envpol.2019.01.067
- Meynecke, J. O., Y. L. Shing, N. C. Duke, and J. Warnken. 2007. Relationships between estuarine habitats and coastal fisheries in Queensland, Australia. Bull. Mar. Sci. **80**: 773–793.
- Moore, C. J., S. L. Moore, M. K. Leecaster, and S. B. Weisberg. 2001. A comparison of plastic and plankton in the North Pacific Central Gyre. Mar. Pollut. Bull. **42**: 1297–1300. doi: 10.1016/S0025-326X(01)00114-X
- Paoli, F., C. S. Wirkner, and S. Cannicci. 2015. The branchiostegal lung of *Uca vocans* (Decapoda: Ocypodidae): Unreported complexity revealed by corrosion casting and MicroCT techniques. Arthropod. Struct. Dev. **44**: 622–629. doi:10.1016/j.asd.2015.09.006
- Peters, C. A., and S. P. Bratton. 2016. Urbanization is a major influence on microplastic ingestion by sunfish in the Brazos River Basin, Central Texas, USA. Environ. Pollut. 210: 380–387. doi:10.1016/j.envpol.2016.01.018
- Poon, D. Y. N., B. K. K. Chan, and G. A. Williams. 2010. Spatial and temporal variation in diets of the crabs *Metopograpsus frontalis* (Grapsidae) and *Perisesarma bidens* (Sesarmidae): Implications for mangrove food webs. Hydrobiologia **638**: 29–40. doi:10.1007/s10750-009-0005-5
- Richards, D. R., and D. A. Friess. 2015. Rates and drivers of mangrove deforestation in Southeast Asia, 2000–2012.
 Proc. Natl. Acad. Sci. USA 113: 344–349. doi:10.1073/pnas. 1510272113

- So, W. K., K. Chan, and C. Not. 2018. Abundance of plastic microbeads in Hong Kong coastal water. Mar. Pollut. Bull. 133: 500–505. doi:10.1016/j.marpolbul.2018.05.066
- Tam, N. F. Y., Y.-S. Wong, C. Y. Lu, and R. Berry. 1997. Mapping and characterization of mangrove plant communities in Hong Kong. Hydrobiologia **352**: 25–37. doi:10.1023/ A:1003032719264
- Tsang, Y. Y., C. W. Mak, C. Liebich, S. W. Lam, E. T. P. Sze, and K. M. Chan. 2017. Microplastic pollution in the marine waters and sediments of Hong Kong. Mar. Pollut. Bull. 115: 20–28. doi:10.1016/j.marpolbul.2016.11.003
- Watts, A. J. R., M. A. Urbina, S. Corr, C. Lewis, and T. S. Galloway. 2015. Ingestion of plastic microfibers by the crab *Carcinus maenas* and its effect on food consumption and energy balance. Environ. Sci. Technol. **49**: 14597–14604. doi:10.1021/acs.est.5b04026
- Watts, A. J. R., M. A. Urbina, R. Goodhead, J. Moger, C. Lewis, and T. S. Galloway. 2016. Effect of microplastic on the gills of the shore crab *Carcinus maenas*. Environ. Sci. Technol. 50: 5364–5369. doi:10.1021/acs.est.6b01187
- Welden, N. A., B. Abylkhani, and L. M. Howarth. 2018. The effects of trophic transfer and environmental factors on microplastic uptake by plaice, *Pleuronectes plastessa*, and spider crab, *Maja squinado*. Environ. Pollut. **239**: 351–358. doi:10.1016/j.envpol.2018.03.110
- Widmer, W. M., and M. C. Hennemann. 2010. Marine debris in the Island of Santa Catarina, South Brazil: Spatial patterns, composition, and biological aspects. J. Coast. Res. 26: 993–1000. doi:10.2112/JCOASTRES-D-09-00072.1
- Wójcik-Fudalewska, D., M. Normant-Saremba, and P. Anastácio. 2016. Occurrence of plastic debris in the stomach of the invasive crab *Eriocheir sinensis*. Mar. Pollut. Bull. **113**: 306–311. doi:10.1016/j.marpolbul.2016.09.059

Acknowledgments

This research was supported by the start-up fund of the University of Hong Kong awarded to C.N. and the seed fund for basic research from the University of Hong Kong awarded to S.C. We would like to thank Laura Elisabeth Agusto and Rebekah Butler for the sampling of the crabs and also Wing Kwan Mandy So for her assistance in laboratory procedures and FTIR analyses.

Submitted 14 March 2019 Revised 12 December 2019 Accepted 19 December 2019